A Randomized Trial of Valganciclovir Prophylaxis Versus Preemptive Therapy in Kidney Transplant Recipients

Tomas Reischig (),^{1,2} Tomas Vlas,^{2,3} Martin Kacer,^{1,2} Kristyna Pivovarcikova,⁴ Daniel Lysak (),^{2,5} Jana Nemcova,^{2,6} Petr Drenko,¹ Jana Machova,^{1,2} Mirko Bouda,^{1,2} Monika Sedivcova,⁶ and Stanislav Kormunda^{2,7}

Due to the number of contributing authors, the affiliations are listed at the end of this article.

ABSTRACT

Background The optimal regimen for preventing cytomegalovirus (CMV) infection in kidney transplant recipients, primarily in reducing indirect CMV effects, has not been defined.

Methods This open-label, single-center, randomized clinical trial of valganciclovir prophylaxis versus preemptive therapy included kidney transplant recipients recruited between June 2013 and May 2018. After excluding CMV-seronegative recipients with transplants from seronegative donors, we randomized 140 participants 1:1 to receive valganciclovir prophylaxis (900 mg, daily for 3 or 6 months for CMV-seronegative recipients who received a kidney from a CMV-seropositive donor) or preemptive therapy (valganciclovir, 900 mg, twice daily) that was initiated after detection of CMV DNA in whole blood (≥1000 IU/ml) and stopped after two consecutive negative tests (preemptive therapy patients received weekly CMV PCR tests for 4 months). The primary outcome was the incidence of biopsy-confirmed acute rejection at 12 months. Key secondary outcomes included subclinical rejection, CMV disease and DNAemia, and neutropenia.

Results The incidence of acute rejection was lower with valganciclovir prophylaxis than with preemptive therapy (13%, 9/70 versus 23%, 16/70), but the difference was not statistically significant. Subclinical rejection at 3 months was lower in the prophylaxis group (13% versus 29%, P = 0.027). Both regimens prevented CMV disease (in 4% of patients in both groups). Compared with prophylaxis, preemptive therapy resulted in significantly higher rates of CMV DNAemia (44% versus 75%, P < 0.001) and a higher proportion of patients experiencing episodes with higher viral load (\geq 2000 IU/ml), but significantly lower valganciclovir exposure and neutropenia.

Conclusion Among kidney transplant recipients, the use of valganciclovir prophylaxis did not result in a significantly lower incidence of acute rejection compared with the use of preemptive therapy.

Clinical Trial Registry Name and Registration Number Optimizing Valganciclovir Efficacy in Renal Transplantation (OVERT Study), ACTRN12613000554763.

JASN 34: 920-934, 2023. doi: https://doi.org/10.1681/ASN.00000000000000000

INTRODUCTION

Cytomegalovirus (CMV) infection belongs to the most common opportunistic infections after organ transplantation. Major advances in the diagnosis and treatment of CMV disease have made it possible to markedly reduce the mortality rates.¹ However, an outstanding issue to be yet tackled is the immunomodulatory properties of CMV associated with a wide range of indirect effects. CMV infection predisposes patients to invasive bacterial or fungal

Received: August 27, 2022. Accepted: December 26, 2022. Published Online Ahead of Print: February 2, 2023.

Correspondence: Prof. Tomas Reischig, Department of Internal Medicine I, Teaching Hospital, alej Svobody 80, 304 60 Pilsen, Czech Republic. Email: reischig@fnplzen.cz

Copyright © 2023 by the American Society of Nephrology

infections and reactivation of other viruses.² CMV contributes to the higher incidence of cardiovascular disease and thrombotic events.^{3,4} Critical for transplant outcomes is its ability to enhance the alloimmune response associated with an increased risk of T-cell–mediated and, possibly, also antibody-mediated allograft rejection, both of which limit long-term graft survival.^{5–8} Even in the modern era with routine use of preventive regimens, CMV infection in kidney transplant recipients entails an increase in allograft failure rates and mortality.^{9,10}

The two main strategies for CMV prevention include (1) universal prophylaxis with treatment of all at-risk patients with antiviral medication and (2) preemptive therapy on the basis of monitoring for CMV and treatment of only selected patients with significant viral replication.^{1,11–17} While prophylaxis with valacyclovir is an option after kidney transplantation, the high pill burden and higher incidence of graft fibrosis makes most centers prefer valganciclovir.12,13,15 The current International Consensus Guidelines consider both strategies for patients after kidney transplantation equivalent, even so for the high-risk subgroup of seronegative recipients of grafts from seropositive donors (D+R-).1 Universal prophylaxis and preemptive therapy differ widely by definition. Preemptive therapy allows the development of early asymptomatic CMV viremia with potential occurrence of indirect CMV effects such as acute rejection.¹⁴ At the same time, CMV viremia episodes controlled by preemptive therapy help boost CMV-specific immunity in D+R- patients and prevent the development of CMV disease in the late post-transplant period.¹⁸ On the other hand, late-onset CMV disease and the myelotoxicity associated with universal prophylaxis entail adverse outcomes.^{11,19,20} It is unclear whether the two strategies differ in the development of ganciclovir resistance.²¹⁻²³

Only few smaller randomized trials with kidney transplant recipients directly comparing valganciclovir prophylaxis and preemptive therapy have been published to date.^{16,17,24} However, two of them used, in their preemptive groups, an insufficiently intensive CMV surveillance monitoring protocol not consistent with current recommendations.^{1,16,17} The remaining study, primarily a pharmacoeconomic one, did not focus on comparing indirect CMV effects.²⁴ Our randomized trial was designed to compare valganciclovir prophylaxis versus preemptive therapy with the primary outcome being the incidence of allograft rejection determined using protocol biopsies and by monitoring of *de novo* donor-specific anti-HLA antibodies (dnDSAs). The trial included a detailed analysis of CMV disease and CMV DNAemia, ganciclovir resistance, CMV-specific T-cell immunity, other infections, and a description of the logistic aspects central to preemptive therapy administration.

METHODS

Study Design

This was an open-label, single-center, randomized clinical trial of valganciclovir prophylaxis versus preemptive therapy in

Significance Statement

Although cytomegalovirus (CMV) infection is an important factor in the pathogenesis of kidney allograft rejection, previous studies have not determined the optimal CMV prevention strategy to avoid indirect effects of the virus. In this randomized trial involving 140 kidney transplant recipients, incidence of acute rejection at 12 months was not lower with valganciclovir prophylaxis (for at least 3 months) compared with preemptive therapy initiated after detection of CMV DNA in whole blood. However, prophylaxis was associated with a lower risk of subclinical rejection at 3 months. Although both regimens were effective in preventing CMV disease, the incidence of CMV DNAemia (including episodes with higher viral loads) was significantly higher with preemptive therapy. Further research with long-term follow-up is warranted to better compare the two approaches.

adult kidney transplant recipients at risk of CMV as defined by donor and recipient CMV serologic combinations of D+R-, D+R+, and D-R+ and recruited between June 2013 and May 2018. Exclusion criteria were participation in another clinical trial, D-R- serostatus, allergy to (val)ganciclovir, severe leukopenia or thrombocytopenia, and inability to provide informed consent. The study was approved by the local ethics committee and conducted in compliance with the Declaration of Helsinki and the Declaration of Istanbul on Organ Trafficking and Transplant Tourism. Written informed consent was obtained before enrollment. The trial was registered with the Australian New Zealand Clinical Trial Registry (ACTRN12613000554763) on May 16, 2023.

Randomization and Interventions

Eligible patients were randomized before transplantation by the transplant physician in a 1:1 ratio to valganciclovir prophylaxis or preemptive therapy using a random-number table with permuted blocks of four. Randomization was stratified by D/R CMV serostatus. Sequentially numbered sealed envelopes were used for allocation concealment. Patients in the prophylaxis group received valganciclovir (Valcyte, Hoffman-La Roche, Grenzach-Wyhlen, Germany) at a dose of 900 mg daily for 3 months (or 6 months in the case of D+R- status) beginning from day 7 after transplant at the latest. Patients in the preemptive group underwent weekly PCR testing for CMV DNA from whole blood for 4 months after transplantation and monthly until 12 months thereafter. PCR was also performed weekly if CMV DNAemia was detected and during the course of preemptive valganciclovir therapy. On detection of CMV DNAemia of \geq 1000 IU/ml, valganciclovir, 900 mg twice daily, was administered within 7 days until two consecutive negative PCR tests a week apart. Recurrent CMV DNAemia was treated similar to the initial episode. In both groups, antiviral drug doses were tapered on the basis of renal function according to the manufacturers' instructions. In patients with antirejection therapy with lymphocyte-depleting antibodies and/or treatment of antibody-mediated rejection after scheduled prophylaxis or weekly PCR testing had been completed, an additional 1-month course of valganciclovir prophylaxis was administered or weekly PCR testing restarted for 1 month, respectively. In the prophylaxis group, the same schedule for PCR CMV DNA testing was applied including weekly testing for the first 4 months; however, asymptomatic CMV DNAemia episodes were not treated.

Immunosuppression and Other Measures of Antimicrobial Prevention

The standard immunosuppressive protocol included tacrolimus, mycophenolate mofetil, and corticosteroids. Immunologically high-risk patients (retransplantation, panel reactive antibody >60%, and/or preformed DSA) received induction by antithymocyte globulin (Thymoglobuline, Genzyme, Lyon, France). Patients after a previous transplantation were tested for the presence of human histocompatibility complex (HLA) antibodies while on the waiting list, and those with preformed DSA underwent desensitization with plasmapheresis and lowdose intravenous immunoglobulin started immediately before transplantation. Desensitization in HLA-incompatible or AB0incompatible living donor transplantation involved anti-CD20 monoclonal antibody (rituximab) and immunoadsorption using staphylococcus protein A columns (Immunosorba, Fresenius Medical Care, Bad Homburg, Germany). Low-dose tacrolimus with anti-IL2R monoclonal antibody (basiliximab) induction was given in recipients of grafts from highly marginal donors (donation after circulatory death, 70 years or older, and/or donors with hypertension or diabetes and significant nephrosclerosis on biopsy). Patients received prophylaxis with trimethoprim-sulfamethoxazole for 4 months and oral amphotericin solution for 1 month. Plasma was tested for polyomavirus BK (BKV) DNA every month for the first 6 months and at 9 and 12 months with preemptive immunosuppression reduction at a significant viral load (≥ 1000 copies/ml).

Outcomes

The primary outcome was the incidence of acute rejection (grade \geq IA or antibody-mediated rejection) at 12 months after transplantation diagnosed by "for-cause" biopsy using the Banff classification.²⁵ Secondary outcomes included CMV DNAemia and CMV disease, CMV ganciclovir resistance, CMV-specific T-cell immunity, subclinical rejection assessed by protocol biopsy at 3 months, development of dnDSA, renal function and graft survival, incidence of other infections, neutropenia, cardiovascular events, malignancy, mortality, and other safety data evaluated by recording adverse events and routine laboratory parameters. CMV DNAemia was defined by the detection of CMV DNA. CMV disease was defined as symptomatic CMV DNAemia and included both CMV syndrome and tissue-invasive disease.^{1,26} All patients remained on follow-up for a minimum of 12 months or until death.

Nucleic Acid Testing for Viral Detection

Quantitative real-time PCR for CMV DNA was performed using a commercially available kit (RealStar CMV PCR kit 1.0, Altona Diagnostics, Hamburg, Germany) according to the manufacturer's instructions on QuantStudio5 (Applied Biosystems, Waltham, MA). DNA was isolated from 200 μ l of whole blood using a commercially available kit (QuickGene DNA whole blood kit S, Kurabo, Japan). Elution was performed in 100 μ l of cell dissociation buffer; the final DNA volume used in the calculation of CMV IU/ml was 100 µl. Quantification was performed using the calibration curve generated from concomitantly amplified quantification standards (IU/ μ l DNA). The lower detection limit of the investigation was 50 IU/ml of whole blood. Quantitative PCR for BKV DNA in the plasma was performed using a commercially available BK Virus R-GENE kit (bioMérieux, Marcy-l'Étoile, France) on a Rotor Gene Q device, according to the manufacturer's instructions (Qiagen, Hilden, Germany) as described previously.^{27,28} The detection limit of BKV DNA was 50 copies/ml. The diagnosis of polyomavirusassociated nephropathy (PVAN) was performed on the basis of both definitive and presumptive PVANs with highgrade BKV viremia (≥10,000 copies/ml) and a negative biopsy. Biopsy-proven definitive PVAN was based on cytopathic changes, confirmed by positive SV40T immunochemistry, and classified according to the new Banff Working Group Classification.²⁹

Analysis of Resistance-Associated Mutations in the UL54 and UL97 Genes

Genotypic testing was performed in specimens with viral loads ≥ 1000 IU/ml as recommended.³⁰ A predefined group was selected for ganciclovir resistance investigation, which included patients with (1) CMV DNAemia during valganciclovir prophylaxis or after prophylaxis completion, (2) persistent CMV DNAemia after at least 3 weeks of preemptive valganciclovir therapy, (3) recurrent CMV DNAemia after previous preemptive valganciclovir therapy, and (4) CMV disease with persistent DNAemia after at least 3 weeks of (val)ganciclovir therapy. The analysis of resistance-associated mutations of CMV (nomenclature NC 006273) was performed using PCR and Sanger sequencing. Investigated regions of the UL54 gene included codons 340-643 and 660-1010 and of UL97 gene codon 434–630. Briefly, 5–10 μ l of isolated DNA was added to a reaction mixture consisting of 12.5 µl of HotStarTaq DNA Polymerase (Qiagen, Hilden, Germany), 10 pmol of forward and reverse primers (Supplemental Table 1), and distilled water up to 25 μ l. The amplification program comprised denaturation at 95°C/14 min, 43× (95°C/40s, 68°C/50s, 72°C/1 min), 72°C/ 7 min, 4°C/(UL54), and 95°C/14 min, 43× (95°C/30s, 68°C/ 30s, 72°C/1 min), 72°C/7 min, 4°C/∞ (UL97). Successfully amplified PCR products selected for sequencing analysis were sequenced on both sides using the Big Dye Terminator Sequencing kit 1.1 (Applied Biosystems, Waltham, MA), according to the manufacturer's protocol. Resistance-associated mutations of CMV were identified using the mutation resistance analyzer database, a web-based search tool that links the sequence to a database containing all published UL97 (protein kinase) and UL54 (DNA polymerase) mutations and corresponding antiviral

www.jasn.org CLINICAL RESEARCH

drug susceptibility phenotypes (https://www.informatik.uniulm.de/ni/mitarbeiter/HKestler/hcmv/).³¹

CMV-Specific T-Cell Immunity

CMV-specific T-cell responses were measured in a cohort of patients (n=95) enrolled until October 2016 before transplantation and at 1, 3, 6, and 12 months thereafter. Quantitation of IFN- γ - and IL-2-producing cells was performed on isolated PBMCs using a modified, dual-color, enzyme-linked immunosorbent spot assay (ELISpot) analysis, as per the manufacturer's instructions (Mabtech, Nacka Strand, Sweden) and as previously described.32,33 PBMCs were obtained using density gradient centrifugation and stored in liquid nitrogen. Thawed PBMCs at concentrations of 2×10^5 cells per well were pipetted into ethanol-treated polyvinylidene difluoride 96-well microplate precoated with primary monoclonal antibodies specific for human IFN- γ and IL-2 (Mabtech, Nacka Strand, Sweden) and stimulated with antigens; an anti-CD3 monoclonal antibody (Mabtech, Nacka Strand, Sweden) and medium alone were used as positive and negative controls, respectively. Two commercially available sets of CMV peptide pools were used for antigenic stimulation: PepMix human cytomegalovirus antigens (pp65) and PepMix human cytomegalovirus antigens (IE-1) (JPT, Berlin, Germany). All stimulations were performed in duplicate wells. Cell cultures were cultivated in a CO₂ incubator overnight, and thereafter, the plates were washed, dried properly, and incubated with detection

antibodies and two fluorescent-labeled conjugates. Fluorescent spots were counted using an automated ELISpot reader equipped with filters for IFN- γ and IL-2 detection (AID, Strassberg, Germany). The mean spot count observed in negative controls was subtracted from the count in the stimulated wells in each run. Results were expressed as the number of spot-forming cells (SFCs) per 2×10⁵ PBMCs.

Assessment of Anti-HLA Antibodies

Donor and recipient HLA typing involved HLA class I (A, B) and class II (DRB1, DQB1). Screening for circulating anti-HLA class I and II antibodies was performed in serum samples at transplantation, 3 and 12 months after transplantation, and, if clinically indicated and determined, using single-antigen flow beads assays on a Luminex platform (Luminex 100/200 System, United States) with the use of commercially available kits (Lifecodes LSA Class I; Lifecodes LSA Class II; Immucor GTI Diagnostic, United States). The results were expressed as mean fluorescence intensity for each HLA specificity. All beads showing a normalized mean fluorescence intensity of >1000 were considered positive.

Statistical Analysis

The null hypothesis was that the acute rejection rate would be equal in the two groups. The null hypothesis would be rejected if valganciclovir prophylaxis was superior to preemptive therapy at the 0.05 level. On the basis of our previous studies and



Figure 1. Flow of patients through the study. CMV, cytomegalovirus; D, donor; R, recipient.

given the use of a tacrolimus-based regimen in all patients, it was estimated that acute rejection would occur in 10% of the patients in the prophylaxis group and 25% of the patients in the preemptive therapy group.^{12,14} A sample size of 90 patients (45 per group) would detect this difference with 80% power and $\alpha = 0.05$. Allowing for dropouts, a minimum of 92 patients were thus planned to enroll. Owing to the clinically important trend toward a lower acute rejection rate in the prophylaxis group (13% versus 27%; P=0.07 by the log-rank test) in interim analysis of the planned study population (n=95), it was decided to continue recruiting patients. Assuming the same difference in acute rejection, at least 134 patients would have been needed. Finally, 140 subjects were planned to enroll anticipating dropouts. In view of the numerically lower incidence of acute rejection in the prophylaxis group, a post hoc analysis according to the type of induction therapy was performed. Quantitative data were compared using the Mann-Whitney U-test or Student t-test as appropriate. Categorical variables were compared using the χ^2 or Fisher exact test. The incidence of acute rejection, CMV DNAemia and disease, BKV viremia, and patient and graft survival rates were calculated using the Kaplan-Meier curves,

with the log-rank test used for comparison. Univariate Cox regression was performed to calculate hazard ratio (HR) and 95% confidence interval (CI) for acute rejection and other variables in the prophylaxis group as compared with the preemptive group. Receiver-operating characteristic curves were used to determine the optimal negative predictive value of various CMV-specific immunity assays for the development of significant (>1000 IU/ml) CMV DNAemia. Statistical calculations were made using SAS software (SAS Institute Inc., Cary, NC). Values of P < 0.05 were considered statistically significant.

RESULTS

Participants

Overall, 153 patients had been screened and 146 subsequently randomized; transplantation was not performed in six patients. The intention-to-treat population included 140 patients (valganciclovir prophylaxis, n=70; preemptive therapy, n=70) (Figure 1). The groups were similar in their basic characteristics (Table 1). Among the 140 randomized patients, 19 (14%) were at risk of

Table 1. Baseline characteristics of the study population

Characteristics	Valganciclovir Prophylaxis (n=70)	Preemptive Therapy ($n=70$)	P Value
Recipient			
Age (yr)	52±11	50±12	0.257
Sex (male)	50 (71)	46 (66)	0.446
Cause of kidney disease			0.404
Chronic glomerulonephritis	35 (50)	32 (46)	
Nephrosclerosis	9 (13)	12 (17)	
Polycystic kidney disease	10 (14)	12 (17)	
Diabetic nephropathy	5 (7)	3 (4)	
Other	11 (16)	11 (16)	
Previous transplantation	5 (7)	8 (11)	0.382
Preemptive transplantation	3 (8)	4 (12)	0.494
HLA A, B, DR, DQ mismatches (n)	4.2±1.6	4.3±1.5	0.747
Pretransplant PRA>20%	8 (11)	4 (6)	0.227
Preformed DSA ^a	10 (14)	13 (19)	0.494
CMV serostatus			0.382
D+R-	9 (13)	10 (14)	
D+R+	56 (80)	52 (74)	
D-R+	5 (7)	8 (11)	
Donor			
Age (yr)	50±16	49±15	0.571
Deceased donor	66 (94)	66 (94)	1.000
Donation after cardiac death	5 (7)	8 (11)	0.382
Living donor ^b	4 (6)	4 (6)	
Expanded criteria donor ^c	29 (41)	29 (41)	1.000
Cold ischemia time (h)	17.2±5.4	17.6±6.3	0.742
Primary immunosuppression			
Tacrolimus + MMF + steroids	70 (100)	70 (100)	
Thymoglobulin induction	9 (13)	10 (14)	0.973
Basiliximab induction	23 (33)	24 (34)	0.858
Desensitization pretransplant ^d	4 (6)	5 (7)	0.730

Data are number of patients (percentage) or mean±SD. PRA, panel reactive antibody; DSA, donor-specific anti-HLA antibody; CMV, cytomegalovirus; D, donor; R, recipient; MMF, mycophenolate mofetil.

^aAt transplantation, using single-antigen flow beads assays on a Luminex platform with inclusion of weak positive results (mean fluorescence intensity >500). ^bIncluding two AB0 incompatible transplantations in valganciclovir prophylaxis and one HLA and AB0 incompatible transplantation in the preemptive prophylaxis group. ^cAccording to the UNOS criteria.

^dValganciclovir prophylaxis: plasmapheresis plus low-dose intravenous immunoglobulin in 3 and immunoadsorption plus rituximab in 1; preemptive therapy: plasmapheresis plus low-dose intravenous immunoglobulin in 4 and immunoadsorption plus rituximab in 1.

primary CMV infection. Induction therapy was received by 66 (47%) patients. Maintenance immunosuppression did not differ between the groups (Supplemental Table 2).

Rejection and dnDSA

Although the number of episodes was higher in the preemptive therapy group, the incidence of the primary outcome of acute rejection at 12 months did not differ significantly between the groups (13%, 9/70 versus 23%, 16/70, P=0.112 [HR, 0.52, 95% CI, 0.23 to 1.19]) (Figure 2A). Unlike patients with thymoglobulin or basiliximab induction, not showing significant differences, valganciclovir prophylaxis in patients without induction therapy was associated with a significant drop in acute rejection (11% versus 31%, P=0.032 [HR, 0.31, 95% CI, 0.10 to 0.98]) (Figure 2B, Table 2). The incidence of acute rejection was higher in the subgroup with CMV DNAemia compared with those not showing CMV replication (24% versus 9%, P=0.022). In the prophylaxis group, CMV DNAemia preceded only 2 (22%) episodes of acute rejection in contrast to 7 (44%) episodes in patients with preemptive therapy (Supplemental Figure 1).

Protocol biopsy at 3 months was performed in 134 (96%) patients. Subclinical rejection including borderline changes was less frequent in patients treated with valganciclovir prophylaxis compared with the preemptive therapy (13% versus 29%, P=0.027); a similar trend was observed when comparing the incidence of at least grade IA rejection episodes (1% versus 8%, P=0.088) (Figure 3A). The Banff interstitial inflammation and tubulitis score was lower in the valganciclovir group (P=0.007) (Figure 3B). The groups did not differ in chronic histological damage and other outcome measures, with a typically high incidence of vascular nephrosclerosis due to the high number of expanded criteria donors (Supplemental Table 3). In patients without preformed DSA, the development of dnDSA was comparable in both groups with a cumulative incidence of 12% versus 5% (P=0.364); however, their presence was often only transient with dnDSA persisting in 7% versus 4% of patients at month 12 (P=0.723) (Table 2). The incidence of delayed graft function, renal function, and proteinuria were comparable (Supplemental Table 4). Likewise, there were no differences in graft and patient survival. In the preemptive therapy group, one patient died because of severe necrotizing pancreatitis while another lost the graft because of severe T-cell-mediated acute rejection associated with renal vein thrombosis.

CMV Disease and CMV DNAemia

Overall, CMV disease was not a common occurrence; the condition was diagnosed in only three patients each in either group (4% versus 4%, P=0.974 [HR, 0.97, 95% CI, 0.20 to 4.82]) (Figure 4A). In the prophylaxis group, the diagnosis was late-onset CMV disease as a rule (>3 months after transplantation). Likewise, patients in the preemptive therapy group experienced two episodes of late-onset CMV disease during the period when PCR monitoring was performed at 1-month intervals. The response to (val)ganciclovir was good in all patients with no recurrent CMV disease (Table 3).



Figure 2. Cumulative incidence of acute rejection. Kaplan-Meier curves for rejection-free survival in (A) the entire study population and (B) patients without induction therapy. CI, confidence interval; HR, hazard ratio.

CMV DNAemia was detected in 31 (44%) and 52 (75%, P<0.001 [HR, 0.32, 95% CI, 0.21 to 0.51]) patients in the prophylaxis and preemptive therapy groups, respectively (Figure 4B). The incidence of CMV DNAemia was significantly lower both in the D+R- group (33% versus 90%, P=0.002) and in R+ patients (46% versus 73%, P<0.001) (Figure 4, C and D). The median time to onset of CMV DNAemia was longer in the valganciclovir prophylaxis group (193 versus 32 days, P<0.001). Consistent with this, late-onset CMV DNAemia was detected more often in the prophylaxis group (36% versus 9%, P < 0.001); however, the difference become nonsignificant when counting recurrent episodes also (P = 0.082) (Table 3). Furthermore, valganciclovir prophylaxis resulted in a decrease in the incidence of CMV DNAemia with a significant (≥2000 IU/ml) viral load (21% versus 49%, P < 0.001) and total duration of CMV DNAemia (25 versus 42 days, P=0.030) within the first year after transplant. Recurrent CMV DNAemia was common, especially in patients with preemptive therapy (42% versus 77%, P=0.002). Detailed virologic characteristics are presented in Table 3.

Table 2.	Incidence a	and	characteristics	of	acute	rejection	and	de	novo	donor-s	pecific	anti	bod	lies

Characteristics	Valganciclovir Prophylaxis (n=70)	Preemptive Therapy (n=70)	P Value
Patients with biopsy for a cause	35 (50)	40 (57)	0.397
Acute rejection (grade ≥IA)	9 (13)	16 (23)	0.112ª
No induction therapy	4 (11)	11 (31)	0.032 ^b
Thymoglobulin	1 (11)	2 (20)	0.577
Basiliximab	4 (17)	3 (13)	0.668
Classification according to Banff 2019			
Grade IA	3 (4)	9 (13)	
Grade IB	1 (1)	0 (0)	
Grade IIA	3 (4)	4 (6)	
Grade IIB	0 (0)	1 (1)	
Grade III	1 (1)	0 (0)	
Active antibody-mediated rejection	2 (3)	3 (5)	
Acute rejection (including borderline)	15 (22)	23 (33)	0.117
Depleting ALA for rejection	8 (11)	7 (10)	0.785
Patients without preformed DSA	60	57	
De novo DSA (cumulative incidence)	7 (12)	3 (5)	0.364
Class I	2 (3)	2 (4)	0.648
Class II	5 (8)	2 (4) ^c	0.464
Peak MFI value	2628 (1781–7275)	1822 (1331–7202)	1.000
De novo DSA at month 3	6 (10)	1 (2)	0.136
De novo DSA at month 12	4 (7)	2 (4)	0.723

Data are number of patients (percentage) or median and interquartile range. ALA, antilymphocyte antibody; DSA, donor-specific anti-HLA antibody; MFI, mean fluorescence intensity; HR, hazard ratio; CI, confidence interval.

^aHR, 0.52; 95% Cl, 0.23 to 1.19.

^cDe novo DSA of both classes I and II were detected in one patient.

On the basis of predefined criteria, genotypic testing was performed in 19 (45 samples) and 29 (73 samples) patients in the valganciclovir prophylaxis and preemptive therapy groups, respectively, without detecting a single mutation associated with ganciclovir resistance. All episodes of CMV DNAemia requiring preemptive therapy were successfully treated with valganciclovir resulting in viral clearance.

Logistics of CMV Prevention, Compliance, and Valganciclovir Exposure

Valganciclovir prophylaxis was administered for an average 104 ± 29 days; this number includes 3 (4%) patients receiving an additional 1-month course of prophylaxis after treatment with lymphocyte-depleting antibodies. Protocol violation

occurred in 1 D+R- patient receiving prophylaxis inadvertently for only 3 instead of 6 months as originally scheduled. In the preemptive therapy group, compliance with the CMV surveillance protocol was high, with 1815 of 1825 (99.5%) per protocol PCR tests performed. At least one PCR test was not performed as scheduled in 7 of 70 (10%) patients; none developed CMV disease. Overall, 38 (54%) patients required a preemptive valganciclovir course, a figure consistent with the 73% of patients with detected CMV DNAemia. The median time from detection of CMV DNAemia exceeding a viral load threshold of 1000 IU/ml to preemptive therapy initiation was 1.5 days. An overwhelming majority (29/38, 76%) of patients required additional courses of preemptive valganciclovir because of recurrent CMV DNAemia. Overall, an average of 2.6 ± 1.4 courses per



Figure 3. Protocol biopsy findings at 3 months after transplantation. (A) Incidence of subclinical rejection including the Banff borderline category, and (B) Banff interstitial inflammation and tubulitis scores. Data are percentage or mean±SEM.

^bHR, 0.31; 95% CI, 0.10 to 0.98.



Figure 4. Cumulative incidence of CMV disease and DNAemia. Kaplan-Meier curves for (A) CMV disease-free survival and CMV DNAemia-free survival in (B) the entire study population, (C) D+R- group, and (D) R+ group. CI, confidence interval; CMV, cyto-megalovirus; D, donor; HR, hazard ratio; R, recipient.

patient had to be administered; the number was significantly higher in the D+R- group (P=0.012) (Table 4).

Despite the need for multiple courses of preemptive treatment with valganciclovir in a significant proportion of patients, total exposure to valganciclovir used in CMV prevention was significantly higher with prophylaxis because of both longer duration of therapy (104 ± 29 versus 33 ± 42 days, P<0.001) and total cumulative dose (70.3 ± 26.4 versus 50.1 ± 65.8 g, P<0.001). An exception to this was the D+R- subgroup having received a comparable cumulative dose (Figure 5, A and B, Supplemental Table 5).

Polyomavirus BK and Other Infections

The two groups did not differ in the incidence of BKV viremia (20% versus 20%, P=0.955) and PVAN (6% versus 6%, P = 1.000). The incidence of other viral, bacterial, and fungal infections was likewise comparable, except for the higher incidence of pneumonia in the preemptive therapy group

(1% versus 10%, P=0.029). However, five of seven patients involved early postoperative pneumonia diagnosed within the first week after transplant (Supplemental Table 6).

Adverse Events

Patients in the valganciclovir prophylaxis group presented more often with neutropenia (41% versus 23%, P=0.019); a similar trend was noticed with leukopenia. However, no significant differences were observed in the incidence of severe neutropenia or use of a granulocyte colony-stimulating factor. A summary of selected adverse events is available in Table 5. Because of its myelotoxicity, valganciclovir doses had to be reduced or temporarily discontinued (median 10 days) in 30% of patients with prophylaxis in contrast to no patient receiving preemptive therapy (P<0.001). Nonetheless, in five of the 38 patients on preemptive valganciclovir therapy, mycophenolate mofetil had to be temporarily withdrawn and/or the granulocyte colony-stimulating factor had to be administered to maintain full valganciclovir dose.

Table 3.	Clinical and	virologic	characteristics	of cyto	omegalovirus	disease	and	DNAemia
----------	--------------	-----------	-----------------	---------	--------------	---------	-----	---------

Characteristics	Valganciclovir Prophylaxis (n=70)	Preemptive Therapy (n=70)	P Value
CMV disease ^a	3 (4)	3 (4)	0.974 ^b
D+R-	1 (11) ^c	2 (20)	0.653
R+	2 (3)	1 (2)	0.597
Late-onset CMV disease ^d	3 (100)	2 (67)	_
Time to onset (d)	191 (179–262)	227 (109–327)	0.986
Time to negative PCR after (val)ganciclovir (d)	22 (12–39)	18 (18–20)	0.609
Recurrent CMV disease	0 (0)	0 (0)	_
CMV DNAemia	31 (44)	52 (75)	<0.001 ^e
D+R-	3 (33)	9 (90)	0.002 ^f
R+	28 (46)	43 (73)	< 0.0019
CMV DNAemia≥2000 IU/ml	15 (21)	34 (49)	< 0.001
D+R-	2 (22)	9 (90)	< 0.001
R+	13 (21)	25 (42)	0.004
Late-onset CMV DNAemia (1st episode)	25 (36)	26 (9)	< 0.001
Late-onset CMV DNAemia (including recurrent episodes)	29 (41)	36 (53)	0.082
Time to onset of 1st episode (d)	193 (141–264)	32 (23–52)	< 0.001
Peak viral load (IU/ml)	1400 (350–11,350)	4250 (1200–19,500)	0.117
D+R-	4100 (2100–965,800)	40,100 (19,800–52,450)	0.355
R+	1350 (300–11,050)	2800 (950–10,800)	0.323
Duration of CMV DNAemia (d)	25 (9–52)	42 (21–75)	0.030
Recurrent CMV DNAemia	13 (42)	40 (77)	0.002

Data are number of patients (percentage) or median and interquartile range. CMV, cytomegalovirus; D, donor; R, recipient; IU, international unit; HR, hazard ratio; CI, confidence interval.

^aValganciclovir prophylaxis: CMV syndrome in 2 and CMV colitis in 1; preemptive therapy: CMV syndrome in 1 and CMV colitis in 2.

^bHR, 0.97; 95% CI, 0.20 to 4.82.

^cA D+R- patient nonadherent to study protocol with prophylaxis withdrawal after 3 months instead of 6 months per protocol.

^dDefined by onset >3 months after transplantation.

^dA single case of early-onset CMV disease in the preemptive therapy group comprised one patient with rapid development of symptoms of CMV syndrome 2 days after a positive PCR test.

^eHR, 0.31; 95% CI, 0.21 to 0.51. ^fHR, 0.15; 95% CI, 0.04 to 0.59. ^gHR, 0.36; 95% CI, 0.22 to 0.59.

CMV-Specific T-Cell Responses

Pretransplant pp65-specific IFN- γ -producing cells (cutoff of 198 SFC/2×10⁵ PBMCs) and both pp65-specific and IE-1-specific IL-2-producing cells (cutoff of 22 SFC/2×10⁵ PBMCs) predicted the development of CMV DNAemia

with >1000 IU/ml with a sensitivity and specificity in the ranges of 75%–94% and 35%–57%, respectively (Supplemental Figures 2A and 2B). The CMV-specific T-cell response after transplant in patients on valganciclovir prophylaxis and preemptive therapy was comparable both in the R+ group (Figure 6)

Table 4.	Details of cytomegaloviru	s prevention o	on the basis	of donor	and recipient	serostatus
----------	---------------------------	----------------	--------------	----------	---------------	------------

Characteristics	All Patients	R+ Group	D+R- Group	P Value
Valganciclovir prophylaxis	n=70	n=61	n=9	_
Duration (d)	104±29	94±6	176±20	< 0.001
Dose (mg/d)	682±189	689±185	630±218	0.451
Preemptive therapy	n=70	n=60	n=10	_
Preemptive valganciclovir				
ITT population	38 (54)	30 (50)	8 (80)	0.097
Patients with CMV	38 (73)	30 (70)	8 (89)	0.415
DNAemia				
Preemptive valganciclovir courses per treated patient	2.6±1.4	2.3±1.2	3.9±1.5ª	0.012
Single course	9 (24)	9 (30)	0 (0)	0.153
Two courses	13 (34)	11 (37)	2 (25)	0.689
Three or more courses	16 (42)	10 (33)	6 (75)	0.050
Duration (d)	69±40	50±35	101±28	0.001
Dose (mg/d)	1523±391	1487±413	1659±279	0.216
Time to initiation (d) ^b	1.9±1.6	1.9±1.7	2.0±1.5	0.338
Viral load at start (IU/ml, median, interquartile range)	3200 (1800–7100)	2800 (1750–6850)	4500 (2500–9400)	0.068

Data are number of patients (percentage) or mean±SD unless otherwise stated. D, donor; R, recipient; ITT, intention-to-treat; CMV, cytomegalovirus; IU, international unit.

^aMaximum of six courses in a single D+R- patient.

^bTime from detection of CMV DNAemia with a viral load exceeding 1000 IU/ml.



Figure 5. Cumulative exposure to valganciclovir for cytomegalovirus prevention in the intention-to-treat population. (A) Duration of therapy, and (B) cumulative valganciclovir dose per patient. Data are mean±SEM. D, donor; R, recipient.

and the D+R- group; however, the number of tested D+R- patients was small (Supplemental Table 7).

DISCUSSION

In our randomized clinical trial with kidney transplant recipients, valganciclovir prophylaxis—compared with preemptive therapy—led to a nonsignificant reduction in the incidence of acute rejection; however, the decrease in patients with a tacrolimus-based regimen without induction therapy was significant at a clinically relevant difference of 20%; in addition, the prophylaxis group showed a lower incidence of subclinical rejection. Although both strategies were effective in reducing the incidence of CMV disease, preemptive therapy was associated with a higher incidence of not only CMV DNAemia but also CMV DNAemia with a higher viral load. A course of preemptive valganciclovir was necessary in half of the patients, with the overwhelming majority requiring multiple courses because of recurrent CMV DNAemia. On the other hand, prophylaxis was associated with a higher incidence of neutropenia resulting in temporary withdrawal of valganciclovir in almost one in four patients.

The most serious indirect effects of CMV include an increased risk of allograft rejection.^{1,5} Regarding the pathogenesis of CMV-associated rejection, several mechanisms have been suggested including stimulation of local inflammation within the allograft with activation of NF- κ B induced, in some patients, by intragraft CMV infection.³⁴ Other hypotheses include heterologous immunity and expansion of natural killer cells, which promote missing self-induced microvascular inflammation, regardless of DSA.6,35,36 The lower incidence of acute rejection and subclinical rejection in patients with valganciclovir prophylaxis is likely in part due to the reduced incidence of CMV DNAemia. However, only approximately half of the rejection episodes were preceded by CMV DNAemia, and the differences seen in the first 2 weeks after transplantation could not be attributed to CMV replication. Another potential factor to be considered is the effect of valganciclovir on T-cell function with an additive immunosuppressive effect.³⁷ This mechanism is supported by an earlier study demonstrating a decrease in the incidence of acute rejection in valganciclovir prophylaxis-as compared with valacyclovir-despite a comparable incidence of CMV DNAemia.¹² Our study documented a comparable incidence of acute rejection in patients with induction therapy, implying a minimal effect of the CMV prevention strategy when using very potent immunosuppression. Compared with preemptive therapy, antiviral prophylaxis in an earlier study with predominantly cyclosporine-based immunosuppression resulted in a lower incidence of rejection.¹⁴ Our study is the first to demonstrate a decreased incidence of rejection when the currently preferred tacrolimus/mycophenolic acid-based protocol is used. Furthermore, this was the first time that a randomized trial documented a beneficial effect of (val)ganciclovir prophylaxis on the incidence of subclinical rejection. Some authors did not demonstrate a significant difference between prophylaxis and preemptive therapy in rejection rates, although in some studies, the incidence was numerically lower with prophylaxis.^{16,17,24} Compared with our study, earlier, patient populations differed not only in immunosuppressive therapy but also in smaller proportions of high-risk expanded-criteria donors. In the US study, almost all patients received thymoglobulin with a very low incidence of acute rejection, whereas in the other studies, there was no clear distribution between the groups in the use of tacrolimus or cyclosporine. Oral ganciclovir prophylaxis used in one study may lead to lower ganciclovir exposure compared with fulldose valganciclovir regimens with a possibly less pronounced effect on T-cell function.¹⁶ A recent multicenter trial showed similar rejection rates in liver transplant recipients.³⁸ In addition to the general difficulty in comparing the risk of rejection between different organs, the significantly higher incidence of CMV disease with prophylaxis could modify CMV indirect effects. It should be noted that the reduction

Table 5. Summary of	of adverse events
---------------------	-------------------

Characteristics	Valganciclovir Prophylaxis (n=70)	Preemptive Therapy (n=70)	P Value
Leukopeniaª	30 (43)	20 (29)	0.078
Neutropeniaª	29 (41)	16 (23)	0.019
Severe neutropeniaª	9 (13)	10 (14)	0.805
Granulocyte colony-stimulating factor use	9 (13)	7 (10)	0.595
Thrombocytopeniaª	11 (16)	8 (11)	0.459
Anemia ^a	6 (9)	12 (17)	0.130
Hallucinations/confusion	4 (6)	5 (7)	0.730
Headache	2 (3)	3 (4)	0.649
Tremor	10 (14)	17 (24)	0.134
Insomnia	9 (13)	8 (11)	0.796
Nausea	20 (29)	22 (31)	0.712
Diarrhea	28 (40)	29 (41)	0.863
New-onset diabetes or IGT ^b	18 (29)	17 (27)	0.799
Hyperlipidemia	59 (84)	53 (76)	0.205
Elevated liver enzymes	14 (20)	22 (31)	0.122
Cardiovascular events	15 (21)	17 (24)	0.687
Malignancy ^c	1 (1)	3 (4)	0.310
Study drug reduction or discontinuation	21 (30)	0 (0) ^d	< 0.001
Reduction	10 (14)	0 (0)	0.014
Discontinuation	16 (23) ^e	0 (0)	0.001

Data are number of patients (percentage). IGT, impaired glucose tolerance.

^aLeukopenia was defined by white blood cell count \leq 3.0 cells/ μ l, neutropenia by absolute neutrophil count \leq 1.5 cells/ μ l with severe neutropenia \leq 0.5 cells/ μ l, thrombocytopenia by \leq 100,000 cells/ μ l, and anemia by hemoglobin level \leq 80 g/L.

^bIn patients without diabetes before transplantation.

^cIncluding nonmelanoma skin cancer.

^dAssessed in 38 patients who received at least one course of preemptive treatment with valganciclovir.

^eReasons for valganciclovir withdrawal: 15 myelotoxicity and one noncompliance.

was observed only with acute T-cell–mediated rejection, not antibody-mediated rejection. This finding is clinically relevant because, even in the era of modern immunosuppression, T-cell–mediated rejection, including subclinical rejection, is associated with a number of adverse outcomes such as persistent inflammation, progression of fibrosis, development of dnDSA and, finally, increased risk of graft loss.^{8,39,40}

In this study and other studies, both strategies of CMV prevention have been shown to be highly effective in preventing the development of CMV disease.^{14,24} Late CMV disease predominated in our study, which may be beneficial given that some authors report its easier control.⁴¹ In the case of preemptive therapy, the prerequisites are weekly monitoring of CMV DNAemia with prompt therapy initiation in patients reaching the viral load threshold and, most importantly, high compliance with the protocol achieved in our study because the preemptive approach may completely fail with a less intense surveillance protocol.^{16,17} Particularly in the preemptive therapy group, we observed a higher incidence of recurrent CMV DNAemia then reported previously.⁴² The finding is likely to be partly influenced by frequent monitoring with high sensitive PCR for CMV. Consistent with earlier studies, no patient with ganciclovir-resistant CMV infection was identified in either group despite systematic genotypic testing^{22,24}; however, our patient population included only a small number of patients at risk of developing ganciclovir resistance.²¹ An important piece of information for patient management was the need to repeat a course of preemptive valganciclovir in most patients because of recurrent CMV DNAemia, a fact not involving only the at-risk D+R- group (although recurrence was clearly the most common issue in this particular subgroup) and one making the logistics of preemptive therapy more complex and intricate. The need to repeat preemptive valganciclovir was partly because of the relatively low viral load threshold (1000 IU/ml) predefined in our study compared with previous trials.14,24 However, the low threshold was chosen to enhance patient safety given the well-known correlation between viral load and CMV disease and, also, the risk of indirect CMV effects.^{10,43} Because of the rapid viral kinetics, the incidence of CMV DNAemia with a higher viral load was even so higher in the preemptive therapy group. It is debatable whether all episodes of asymptomatic CMV DNAemia reaching threshold levels indeed required therapy. It has been suggested that T-cell-mediated CMV-specific immunity be included in the decision-making algorithm of preemptive therapy.44

Despite multiple courses of preemptive valganciclovir, total valganciclovir exposure was markedly lower in patients managed by preemptive therapy with an approximately three times shorter therapy duration and 30% cumulative dose reduction. The lower exposure to valganciclovir is associated with lower costs and, most importantly, lower drug toxicity.¹⁹ Our study documented a higher incidence of leukopenia and neutropenia in the prophylaxis group; valganciclovir had to be temporarily discontinued in 23% of patients. However, no difference was found in the incidence of severe neutropenia or use of the granulocyte colony-stimulating factor. Overall, the safety profile of valganciclovir prophylaxis was a favorable



Figure 6. Cytomegalovirus-specific T-cell response by IFN- γ -secreting cells in seropositive recipients. Assessed by the enzyme-linked immunosorbent spot assay after stimulation by (A) pp65 or (B) IE-1 antigen pools and expressed as SFC per 2×10⁵ PBMCs. No significant differences were found in comparisons between prophylaxis and preemptive therapy. In both valganciclovir prophylaxis and preemptive therapy groups, there were significant differences between pre-transplant and post-transplant values in both pp65-specific (*P*<0.001) and IE-1–specific (*P*=0.001 and *P*=0.004) response by Friedman analysis of variance on ranks. Minimum and maximum values excluding outliers are represented by whiskers; median and interquartile range are inside boxes. SFC, spot-forming cell.

one. The safety of prophylaxis may even be improved by the use of T-cell–mediated CMV-specific immunity to guide early discontinuation of valganciclovir.⁴⁵

Except for the higher incidence of pneumonia in the preemptive therapy group, which, however, was observed in the early postoperative period and is unlikely to be related to CMV prevention, there were no differences between the two groups in the incidence of other infections. The incidence of BKV viremia and PVAN was also identical, although our previous study documented an increased risk associated with valganciclovir prophylaxis with potential impairment of T-cell function, a critical factor in the control of BKV replication.^{27,46} However, over half of the patients in the preemptive therapy group were also treated with valganciclovir, and the higher incidence of acute rejection and antirejection therapy resulted in accumulation of immunosuppression, possibly affecting the incidence of BKV viremia. It should be noted that the increase in BKV viremia was seen in patients receiving valganciclovir prophylaxis only versus valacyclovir prophylaxis and not preemptive therapy.¹² To be able to draw authoritative conclusions, further studies are warranted comparing valganciclovir prophylaxis with other antiviral drugs not affecting T-cell function such as maribavir.⁴⁷

Similar to previous studies, the level of pretransplant CMV-specific T-cell immunity was predictive of the development of CMV DNAemia.^{44,48} CMV immunity after transplantation was comparable in both groups, a surprising observation especially in D+R– patients. When using the preemptive approach, liver transplant recipients have been repeatedly shown to experience more rapid development of CMV-specific T-cell response as a consequence of previous episodes of CMV DNAemia.^{18,38} In our study, ELISpot could be performed in only a small number of D+R– patients, a fact making relevant conclusions impossible.

This study has several limitations. First, for reasons of logistics, given the nature of interventions in the preemptive therapy group, it was not a blinded one. Nonetheless, the main outcomes, i.e., rejection and CMV DNAemia, were defined by the biopsy finding or PCR test, and both the pathologists and the physicians assessing the latter were blinded to treatment allocation. Furthermore, the single-center design precludes generalization of our results to another population, a fact applying primarily to D+R- patients whose number was small and making conclusive analyses impossible. The effect on acute rejection or prevention of CMV may differ with other immunosuppressive protocols, not only generally, when using induction therapy, but also in regimens including a mammalian target of rapamycin inhibitors, associated with a lower risk of CMV infection.⁴⁹ Currently, most kidney transplant recipients, mostly in the United States, receive induction therapy, which limits the potential benefit of valganciclovir prophylaxis.⁵⁰ On the other hand, the reduction in subclinical rejection did not seem to be dependent on induction therapy. Last but not least, a consideration not to be disregarded is the ability of the transplant center to ensure intensive CMV monitoring and patient compliance, as well as a prompt initiation of preemptive valganciclovir therapy, factors critical for the success of preemptive therapy. Annual follow-up and early protocol biopsy do not allow to detect subclinical rejection in the later post-transplant period and are inadequate to assess the real effect of late-onset CMV DNAemia, a condition characteristic for prophylaxis. As a result, short-term outcomes may put the preemptive therapy approach at a disadvantage, as was the case in a previous comparison with valacyclovir prophylaxis.¹⁵ On the other hand, asymptomatic CMV DNAemia after prophylaxis completion was not treated in our study. It cannot be ruled out that possible

treatment could improve the prevention of CMV indirect effects. The planned long-term follow-up and protocol biopsy at 3 years should answer the question whether a decrease in the incidence of acute rejection in the early posttransplant period in patients receiving valganciclovir prophylaxis will alleviate the chronic histologic changes and graft function in the long run.

In conclusion, compared with preemptive therapy, valganciclovir prophylaxis in kidney transplant recipients did not result in a significantly lower incidence of acute rejection. Prophylaxis is associated with a lower risk of subclinical rejection at month 3 and reduced incidence of CMV DNAemia. In the presence of intensive CMV monitoring and maintenance of a high compliance rate, preemptive therapy is equally effective in preventing CMV disease at a lower cumulative valganciclovir exposure and a lower incidence of neutropenia. Further research with long-term follow-up is warranted to be able to finally compare both regimens.

DISCLOSURES

D. Lysak reports Consultancy: AbbVie, AstraZeneca, Janssen, and Novartis. J. Machová reports Other Interests or Relationships: Member of the Czech Society of Nephrology and Member of the Czech Transplant Society. All remaining authors have nothing to disclose.

FUNDING

This work was supported by the Charles University Cooperation Program, research areas "Immunity and Infections" and "Internal Disciplines," and project "Fighting INfectious Diseases" [CZ.02.1.01/0.0/0.0/16_019/0000787] awarded by the Ministry of Education, Youth and Sports of the Czech Republic, financed from The European Regional Development Fund.

ACKNOWLEDGMENTS

We are most grateful to our friend Prof. Ondrej Hes (†2022) for his excellent contribution, as a nephropathologist, to this trial and previous research. We thank Lenka Karlikova and Jana Havlova, the transplant coordinator nurses, for their assistance in data collection.

AUTHOR CONTRIBUTIONS

Conceptualization: T. Reischig Data curation: T. ReischigFormal analysis: S. Kormunda, T. Reischig Funding acquisition: T. Reischig Investigation: M. Bouda, P. Drenko, M. Kacer, D. Lysak, J. Machova, J. Nemcova, K. Pivovarcikova, T. Reischig, M. Sedivcova, T. Vlas Methodology: M. Kacer, S. Kormunda, D. Lysak, J. Nemcova, K. Pivovarcikova, T. Reischig, M. Sedivcova, T. Vlas Project administration: T. Reischig Supervision: T. Reischig Validation: T. Reischig Writing original draft: T. Reischig Writing review & editing: M. Bouda, P. Drenko, M. Kacer, S. Kormunda, D.

Lysak, J. Machova, J. Nemcova, K. Pivovarcikova, T. Reischig, M. Sedivcova, T. Vlas

DATA SHARING STATEMENT

The clinical trial dataset is available through the persistent web link: https://ldrv.ms/u/s!AkD3z7X17uL90kV4sp1EOCGSCyO2?e=nEWH8w

SUPPLEMENTAL MATERIAL

This article contains the following supplemental material online at http://links.lww.com/JSN/D743.

Supplemental Figure 1. Timing of acute rejection and CMV DNAemia in individual patients.

Supplemental Figure 2A and B. Receiver-operating characteristic curve analysis for prediction of CMV DNAemia.

- Supplemental Table 1. PCR and sequencing primers used in analysis of resistance-associated mutations in UL54 and UL97 genes.
 - Supplemental Table 2. Immunosuppressive therapy during the study.

Supplemental Table 3. Protocol biopsy findings at 3 months after transplantation.

Supplemental Table 4. Renal function and graft and patient survival.

Supplemental Table 5. Cumulative valganciclovir exposure for cytomegalovirus prevention in an intention-to-treat population.

Supplemental Table 6. Incidence of infection other than cytomegalovirus within 12 months.

Supplemental Table 7. Cytomegalovirus-specific T-cell responses assessed by ELISpot.

REFERENCES

- Kotton CN, Kumar D, Caliendo AM, Huprikar S, Chou S, Danziger-Isakov L. The third international consensus guidelines on the management of cytomegalovirus in solid-organ transplantation. *Transplantation*. 2018;102(6):900–931. doi:10.1097/tp.00000000002191
- L'Huillier AG, Ferreira VH, Ku T, Bahinskaya I, Kumar D, Humar A. Improving our mechanistic understanding of the indirect effects of CMV infection in transplant recipients. *Am J Transplant.* 2019;19(9): 2495–2504. doi:10.1111/ajt.15371
- Courivaud C, Bamoulid J, Chalopin JM, Gaiffe E, Tiberghien P, Saas P. Cytomegalovirus exposure and cardiovascular disease in kidney transplant recipients. J Infect Dis. 2013;207(10):1569–1575. doi:10. 1093/infdis/jit064
- Belga S, MacDonald C, Chiang D, Kabbani D, Shojai S, Abraldes JG. Donor graft cytomegalovirus serostatus and the risk of arterial and venous thrombotic events in seronegative recipients after non-thoracic solid organ transplantation. *Clin Infect Dis.* 2021;72(5):845–852. doi:10.1093/cid/ciaa125
- Reischig T, Jindra P, Svecova M, Kormunda S, Opatrny K Jr., Treska V. The impact of cytomegalovirus disease and asymptomatic infection on acute renal allograft rejection. *J Clin Virol.* 2006;36(2):146–151. doi:10. 1016/j.jcv.2006.01.015
- Bachelet T, Couzi L, Pitard V, Sicard X, Rigothier C, Lepreux S. Cytomegalovirus-responsive gammadelta T cells: novel effector cells in antibody-mediated kidney allograft microcirculation lesions. J Am Soc Nephrol. 2014;25(11):2471–2482. doi:10.1681/ASN.2013101052
- Loupy A, Lefaucheur C. Antibody-mediated rejection of solid-organ allografts. N Engl J Med. 2018;379(12):1150–1160. doi:10.1056/ NEJMra1802677
- Rampersad C, Balshaw R, Gibson IW, Ho J, Shaw J, Karpinski M. The negative impact of T cell-mediated rejection on renal allograft survival in the modern era. Am J Transplant. 2022;22(3):761–771. doi:10.1111/ajt.16883
- Leeaphom N, Garg N, Thamcharoen N, Khankin EV, Cardarelli F, Pavlakis M. Cytomegalovirus mismatch still negatively affects patient and graft survival in the era of routine prophylactic and preemptive therapy: a paired kidney analysis. *Am J Transplant.* 2019;19(2): 573–584. doi:10.1111/ajt.15183

- Reischig T, Kacer M, Hruba P, Jindra P, Hes O, Lysak D. The impact of viral load and time to onset of cytomegalovirus replication on longterm graft survival after kidney transplantation. *Antivir Ther.* 2017; 22(6):503–513. doi:10.3851/IMP3129
- Humar A, Lebranchu Y, Vincenti F, Blumberg EA, Punch JD, Limaye AP. The efficacy and safety of 200 days valganciclovir cytomegalovirus prophylaxis in high-risk kidney transplant recipients. *Am J Transplant*. 2010;10(5):1228–1237. doi:10.1111/j.1600-6143.2010.03074.x
- Reischig T, Kacer M, Jindra P, Hes O, Lysak D, Bouda M. Randomized trial of valganciclovir versus valacyclovir prophylaxis for prevention of cytomegalovirus in renal transplantation. *Clin J Am Soc Nephrol.* 2015;10(2):294–304. doi:10.2215/CJN.07020714
- Reischig T, Kacer M, Hruba P, Hermanova H, Hes O, Lysak D. Less renal allograft fibrosis with valganciclovir prophylaxis for cytomegalovirus compared to high-dose valacyclovir: a parallel group, open-label, randomized controlled trial. *BMC Infect Dis.* 2018;18(1):573. doi:10. 1186/s12879-018-3493-y
- Reischig T, Jindra P, Hes O, Svecova M, Klaboch J, Treska V. Valacyclovir prophylaxis versus preemptive valganciclovir therapy to prevent cytomegalovirus disease after renal transplantation. *Am J Transplant*. 2008;8(1):69–77. doi:10.1111/j.1600-6143.2007.02031.x
- Reischig T, Hribova P, Jindra P, Hes O, Bouda M, Treska V. Long-term outcomes of pre-emptive valganciclovir compared with valacyclovir prophylaxis for prevention of cytomegalovirus in renal transplantation. J Am Soc Nephrol. 2012;23(9):1588–1597. doi:10.1681/ ASN.2012010100
- Kliem V, Fricke L, Wollbrink T, Burg M, Radermacher J, Rohde F. Improvement in long-term renal graft survival due to CMV prophylaxis with oral ganciclovir: results of a randomized clinical trial. Am J Transplant. 2008;8(5):975–983. doi:10.1111/j.1600-6143.2007.02133.x
- Witzke O, Nitschke M, Bartels M, Wolters H, Wolf G, Reinke P. Valganciclovir prophylaxis versus preemptive therapy in cytomegaloviruspositive renal allograft recipients: long-term results after 7 years of a randomized clinical trial. *Transplantation*. 2018;102(5):876–882. doi: 10.1097/tp.00000000002024
- Limaye AP, Green ML, Edmison BC, Stevens-Ayers T, Chatterton-Kirchmeier S, Geballe AP. Prospective assessment of cytomegalovirus immunity in high-risk donor-seropositive/recipient-seronegative liver transplant recipients receiving either preemptive therapy or antiviral prophylaxis. J Infect Dis. 2019;220(5):752–760. doi:10.1093/infdis/jiz181
- Florescu DF, Qiu F, Schmidt CM, Kalil AC. A direct and indirect comparison meta-analysis on the efficacy of cytomegalovirus preventive strategies in solid organ transplant. *Clin Infect Dis.* 2014;58(6): 785–803. doi:10.1093/cid/cit945
- Dube GK, Morris HK, Crew RJ, Pereira MR, Cohen DJ, Mohan S. Febrile neutropenia after kidney transplantation. *Am J Transplant*. 2021; 21(10):3436–3443. doi:10.1111/ajt.16714
- Fisher CE, Knudsen JL, Lease ED, Jerome KR, Rakita RM, Boeckh M. Risk factors and outcomes of ganciclovir-resistant cytomegalovirus infection in solid organ transplant recipients. *Clin Infect Dis.* 2017;65(1): 57–63. doi:10.1093/cid/cix259
- Boivin G, Goyette N, Farhan M, Ives J, Elston R. Incidence of cytomegalovirus UL97 and UL54 amino acid substitutions detected after 100 or 200 days of valganciclovir prophylaxis. *J Clin Virol.* 2012;53(3): 208–213. doi:10.1016/j.jcv.2011.12.019
- Couzi L, Helou S, Bachelet T, Moreau K, Martin S, Morel D. High incidence of anticytomegalovirus drug resistance Among D+R- kidney transplant recipients receiving preemptive therapy. Am J Transplant. 2012;12(1):202–209. doi:10.1111/j.1600-6143.2011.03766.x
- Khoury JA, Storch GA, Bohl DL, Schuessler RM, Torrence SM, Lockwood M. Prophylactic versus preemptive oral valganciclovir for the management of cytomegalovirus infection in adult renal transplant recipients. Am J Transplant. 2006;6(9):2134–2143. doi:10.1111/j. 1600-6143.2006.01413.x
- Loupy A, Haas M, Roufosse C, Naesens M, Adam B, Afrouzian M. The Banff 2019 Kidney Meeting Report (I): updates on and clarification of

criteria for T cell- and antibody-mediated rejection. *Am J Transplant*. 2020;20(9):2318–2331. doi:10.1111/ajt.15898

- Ljungman P, Boeckh M, Hirsch HH, Josephson F, Lundgren J, Nichols G. Definitions of cytomegalovirus infection and disease in transplant patients for use in clinical trials. *Clin Infect Dis.* 2017;64(1):87–91. doi: 10.1093/cid/ciw668
- Reischig T, Kacer M, Hes O, Machova J, Nemcova J, Lysak D. Cytomegalovirus prevention strategies and the risk of BK polyomavirus viremia and nephropathy. *Am J Transplant*. 2019;19(9):2457–2467. doi:10.1111/ajt.15507
- Reischig T, Kacer M, Hes O, Machova J, Nemcova J, Kormunda S. Viral load and duration of BK polyomavirus viraemia determine renal graft fibrosis progression: histologic evaluation of late protocol biopsies. *Nephrol Dial Transplant*. 2019;34(11):1970–1978. doi:10.1093/ndt/gfz061
- Nickeleit V, Singh HK, Dadhania D, Cornea V, El-Husseini A, Castellanos A. The 2018 Banff Working Group classification of definitive polyomavirus nephropathy: a multicenter validation study in the modern era. *Am J Transplant*. 2021;21(2):669–680. doi:10.1111/ ajt.16189
- Chemaly RF, Chou S, Einsele H, Griffiths P, Avery R, Razonable RR. Definitions of resistant and refractory cytomegalovirus infection and disease in transplant recipients for use in clinical trials. *Clin Infect Dis.* 2019;68(8):1420–1426. doi:10.1093/cid/ciy696
- Chevillotte M, von Einem J, Meier BM, Lin FM, Kestler HA, Mertens T. A new tool linking human cytomegalovirus drug resistance mutations to resistance phenotypes. *Antivir Res.* 2010;85(2):318–327. doi:10. 1016/j.antiviral.2009.10.004
- Korber N, Behrends U, Hapfelmeier A, Protzer U, Bauer T. Validation of an IFNγ/IL2 FluoroSpot assay for clinical trial monitoring. *J Transl Med.* 2016;14(1):175. doi:10.1186/s12967-016-0932-7
- Reischig T, Kacer M, Vlas T, Drenko P, Kielberger L, Machova J. Insufficient response to mRNA SARS-CoV-2 vaccine and high incidence of severe COVID-19 in kidney transplant recipients during pandemic. *Am J Transplant*. 2022;22(3):801–812. doi:10.1111/ajt.16902
- Dzabic M, Rahbar A, Yaiw KC, Naghibi M, Religa P, Fellstrom B. Intragraft cytomegalovirus protein expression is associated with reduced renal allograft survival. *Clin Infect Dis.* 2011;53(10):969–976. doi:10.1093/cid/cir619
- Heutinck KM, Yong SL, Tonneijck L, van den Heuvel H, van der Weerd NC, van der Pant KA. Virus-specific CD8(+) T cells cross-reactive to donor-alloantigen are transiently present in the circulation of kidney transplant recipients infected with CMV and/or EBV. Am J Transplant. 2016;16(5):1480–1491. doi:10.1111/ajt.13618
- Callemeyn J, Senev A, Coemans M, Lerut E, Sprangers B, Kuypers D. Missing self-induced microvascular rejection of kidney allografts: a population-based study. J Am Soc Nephrol. 2021;32(8):2070–2082. doi:10.1681/ASN.2020111558
- Reischig T, Prucha M, Sedlackova L, Lysak D, Jindra P, Bouda M. Valganciclovir prophylaxis against cytomegalovirus impairs lymphocyte proliferation and activation in renal transplant recipients. *Antivir Ther.* 2011;16(8):1227–1235. doi:10.3851/imp1879
- 38. Singh N, Limaye AP. Preemptive therapy vs antiviral prophylaxis in cytomegalovirus-seronegative liver transplant recipients with

seropositive donors-reply. JAMA. 2020;324(11):1108–1109. doi:10. 1001/jama.2020.11865

- El Ters M, Grande JP, Keddis MT, Rodrigo E, Chopra B, Dean PG. Kidney allograft survival after acute rejection, the value of follow-up biopsies. Am J Transplant. 2013;13(9):2334–2341. doi:10.1111/ajt. 12370
- Garcia-Carro C, Dorje C, Asberg A, Midtvedt K, Scott H, Reinholt FP. Inflammation in early kidney allograft surveillance biopsies with and without associated tubulointerstitial chronic damage as a predictor of fibrosis progression and development of de novo donor specific antibodies. *Transplantation*. 2017;101(6):1410–1415. doi:10.1097/tp.000000000001216
- Kaminski H, Couzi L, Garrigue I, Moreau JF, Dechanet-Merville J, Merville P. Easier control of late-onset cytomegalovirus disease following universal prophylaxis through an early antiviral immune response in donor-positive, recipient-negative kidney transplants. Am J Transplant. 2016;16(8):2384–2394. doi:10.1111/ajt.13781
- Kumar D, Mian M, Singer L, Humar A. An interventional study using cell-mediated immunity to personalize therapy for cytomegalovirus infection after transplantation. *Am J Transplant.* 2017;17(9): 2468–2473. doi:10.1111/ajt.14347
- Natori Y, Alghamdi A, Tazari M, Miller V, Husain S, Komatsu T. Use of viral load as a surrogate marker in clinical studies of cytomegalovirus in solid organ transplantation: a systematic review and meta-analysis. *Clin Infect Dis.* 2018;66(4):617–631. doi:10.1093/cid/cix793
- Jarque M, Crespo E, Melilli E, Gutierrez A, Moreso F, Guirado L. Cellular immunity to predict the risk of cytomegalovirus infection in kidney transplantation: a prospective, interventional, multicenter clinical trial. *Clin Infect Dis.* 2020;71(9):2375–2385. doi:10.1093/cid/ ciz1209
- Paez-Vega A, Gutierrez-Gutierrez B, Aguera ML, Facundo C, Redondo-Pachon D, Suner M. Immunoguided discontinuation of prophylaxis for cytomegalovirus disease in kidney transplant recipients treated with antithymocyte globulin: a randomized clinical trial. *Clin Infect Dis.* 2022;74(5):757–765. doi:10.1093/cid/ciab574
- 46. Schachtner T, Stein M, Babel N, Reinke P. The loss of BKV-specific immunity from pretransplantation to posttransplantation identifies kidney transplant recipients at increased risk of BKV replication. Am J Transplant. 2015;15(8):2159–2169. doi:10.1111/ajt.13252
- Stachel D, Stevens-Ayers T, Boeckh M. In vitro studies of the impact of maribavir on CMV-specific cellular immune responses. J Clin Virol. 2016;75:53–59. doi:10.1016/j.jcv.2015.12.012
- Kumar D, Chin-Hong P, Kayler L, Wojciechowski D, Limaye AP, Osama Gaber A. A prospective multicenter observational study of cell-mediated immunity as a predictor for cytomegalovirus infection in kidney transplant recipients. Am J Transplant. 2019;19(9):2505–2516. doi:10.1111/ajt.15315
- 49. Kaminski H, Marseres G, Yared N, Nokin MJ, Pitard V, Zouine A. mTOR inhibitors prevent CMV infection through the restoration of functional $\alpha\beta$ and $\gamma\delta$ T cells in kidney transplantation. *J Am Soc Nephrol.* 2022; 33(1):121–137. doi:10.1681/ASN.2020121753
- Lentine KL, Smith JM, Hart A, Miller J, Skeans MA, Larkin L. OPTN/SRTR 2020 annual data report: kidney. Am J Transplant. 2022; 22(suppl 2):21–136. doi:10.1111/ajt.16982

AFFILIATIONS

¹Department of Internal Medicine I, Faculty of Medicine in Pilsen, Charles University, and Teaching Hospital, Pilsen, Czech Republic

²Biomedical Centre, Faculty of Medicine in Pilsen, Charles University, Pilsen, Czech Republic

³Department of Immunology and Allergology, Faculty of Medicine in Pilsen, Charles University, and Teaching Hospital, Pilsen, Czech Republic ⁴Department of Pathology, Faculty of Medicine in Pilsen, Charles University, and Teaching Hospital, Pilsen, Czech Republic

⁵Department of Hematology and Oncology, Faculty of Medicine in Pilsen, Charles University, and Teaching Hospital, Pilsen, Czech Republic ⁶Department of Molecular Genetic, Biopticka laboratory, Pilsen, Czech Republic

⁷Division of Information Technologies and Statistics, Faculty of Medicine in Pilsen, Charles University, Pilsen, Czech Republic